Author's response to reviews

Title: Meta-analysis of randomised Phase II trials to inform subsequent Phase III decisions

Authors:

Danielle L Burke (dlb735@bham.ac.uk)
Lucinda J Billingham (l.j.billingham@bham.ac.uk)
Alan J Girling (a.j.girling@bham.ac.uk)
Richard D Riley (r.d.riley@bham.ac.uk)

Version: 3
Date: 11 August 2014

Author's response to reviews: see over
Response to Editor Comments

Ref: 1581369937124657
Authors: Burke, Billingham, Girling, Riley
Title: Meta-analysis of randomised Phase II trials to inform subsequent Phase III decisions

We would like to thank the editor for accepting the revised manuscript. A response to the editor’s formatting requests is given below. The editor comments are shown in bold italics, and our responses are in normal text.

RESPONSE TO EDITORIAL REQUESTS

1. Please include a Competing Interests section at the end of the manuscript, before the Reference List. If the authors have no competing interests, please state: "The authors declare that they have no competing interests."

   Added the statement: “The authors declare that they have no competing interests” before the Reference List.

2. Please include a list of abbreviations used and their meanings.

   Manuscript now includes a list of abbreviations and their meanings after the Conclusions section.